
Daily status, October 17
10/17/13

Summary:
Hardwired floor radiation and soil to the DSM yesterday.
IOP 3 last night.

T/RH: RH.40m.near and RH.35m.rim perhaps a bit high.
         Ifan suspicious at near.30m and T a bit high both day and night.
P: P.nne down 10/16 09:45 - 12:10
csat u,v: ok
csat ldiag: ok
csat w, tc: ok
fluxes: ok
kh2o: FLR variance high at night, but also tc'tc'
motes: FLR OK! after 14:00 10/16
           FAR soil mote stopped 10/16 13:45 - 15:40.
           NEAR ok
Wetness: ok
radiation: ok
Tsoil: Tsoil.2.5cm.flr & far (linear avg) do not fit profile
Gsoil: ok
Qsoil: ok
Cvsoil: ok
2D sonic: ok
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